Author's response to reviews

Title: Is there a difference? Patients with chronic widespread pain and chronic local back pain in primary care - A comparative cross-sectional analysis

Authors:

Annika Viniol (annika.viniol@staff.uni-marburg.de)
Nikita Jegan (jegan@staff.uni-marburg.de)
Corinna Leonhardt (leonhar3@staff.uni-marburg.de)
Markus Brugger (markus.brugger@helmholtz-muenchen.de)
Konstantin Strauch (strauch@helmholtz-muenchen.de)
Jürgen Barth (mail@juergen-barth.de)
Erika Baum (erika.baum@staff.uni-marburg.de)
Annette Becker (annette.becker@staff.uni-marburg.de)

Version: 4 Date: 27 November 2013

Author's response to reviews: see over
Author's response to reviews

Title: Is there a difference? Patients with chronic widespread pain and chronic local back pain in primary care - A comparative cross-sectional analysis

Authors:
Annika Viniol: annika.viniol@staff.uni-marburg.de
Nikita Jegan: jegan@staff.uni-marburg.de
Corinna Leonhardt: leonhar3@staff.uni-marburg.de
Markus Brugger: markus.brugger@helmholtz-muenchen.de
Konstantin Strauch: strauch@helmholtz-muenchen.de
Jürgen Barth: mail@juergen-barth.de
Erika Baum: erika.baum@staff.uni-marburg.de
Annette Becker: annette.becker@staff.uni-marburg.de

Version 3: Date: 27 November 2013

Author's response to reviews: see over
Reviewer's report

Title: Is there a difference? Patients with chronic widespread pain and chronic local back pain in primary care - A comparative cross-sectional analysis

Version: 3 Date: 18 November 2013

Reviewer: Bård Natvig

Reviewer's report:
The article is definitely improved through the revision process. In my opinion it is now acceptable for publication.

I have some tiny remarks:
page 6, line 9. ref for Wolfe et al should be 1 (and not 24)
page 8, line 18 and 19. drop "was" and "of"?
page 12, line 17. a reference to von Korff should be added.

Level of interest: An article of importance in its field

Quality of written English: Acceptable

Statistical review: No, the manuscript does not need to be seen by a statistician.

Declaration of competing interests:
I declare that I have no competing interests

I included all remarks. Thank you for the review.
Reviewer's report

Title: Is there a difference? Patients with chronic widespread pain and chronic local back pain in primary care - A comparative cross-sectional analysis

Version: 3 Date: 7 November 2013

Reviewer: Lise Hestbaek

Reviewer's report: I am satisfied with the provided changes and no further comments

Level of interest: An article whose findings are important to those with closely related research interests

Quality of written English: Acceptable

Statistical review: No, the manuscript does not need to be seen by a statistician.

Declaration of competing interests: I have no competing interests

Thank you for the review.